Author's response to reviews

Title: Homebirth and barriers to referring women with obstetric complications to hospitals: a mixed methods research in Zahedan, Southeast of Iran

Authors:

Mahmoud Ghazi Tabatabaie (smghazi@ut.ac.ir)
Zahra Moudi (moudi@zaums.ac.ir)
Abouali Vedadh (vedadha@ut.ac.ir)

Version: 2 Date: 20 February 2012

Author's response to reviews:

Dear Sir/Madam

Reproductive Health Journal

We are enclosing herewith a manuscript entitled “Homebirth and barriers to referring women with obstetric complications to hospitals: a mixed methods research in Zahedan, Southeast of Iran” submitted to “Reproductive Health Journal” for possible review and publishing. With the submission of this manuscript we would like to undertake that our joint manuscript has not been published elsewhere, accepted for publication elsewhere or under editorial review for publication elsewhere.

This is an original article and includes rich evidence and discussions about the subject. Considering the vision and mission of the journal, we believe that our joint article is closely relevant to subject area of the world-class journal of Reproductive Health.

Thank you in advance for your time and reviewing the manuscript,

Sincerely yours,

The Authors